Check for updates

OPEN ACCESS

EDITED BY Vsevolod Konstantinov, Penza State University, Russia

REVIEWED BY

Dhirendra Paudel, Mental Health and Yoga Pvt Ltd., Nepal Giorgia Varallo, University of Parma, Italy

*CORRESPONDENCE Dan Shan D.Shan6@nuigalway.ie

[†]These authors have contributed equally to this work

SPECIALTY SECTION

This article was submitted to Anxiety and Stress Disorders, a section of the journal Frontiers in Psychiatry

RECEIVED 24 June 2022 ACCEPTED 16 August 2022 PUBLISHED 30 August 2022

CITATION

Shan D, Liu C, Li S and Zheng Y (2022) Increased anxiety from fear of Omicron in China as compared to North America and Western Europe: A cross-sectional Kendall's tau-b analysis using the generalized anxiety disorder 7-item questionnaire. *Front. Psychiatry* 13:977361. doi: 10.3389/fpsyt.2022.977361

COPYRIGHT

© 2022 Shan, Liu, Li and Zheng. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Increased anxiety from fear of Omicron in China as compared to North America and Western Europe: A cross-sectional Kendall's tau-b analysis using the generalized anxiety disorder 7-item questionnaire

Dan Shan^{1*†}, Chang Liu^{2†}, Shaoyang Li^{3†} and Yuandian Zheng⁴

¹School of Medicine, National University of Ireland, Galway, Ireland, ²School of Clinical Medicine, University of Cambridge, Cambridge, United Kingdom, ³Faculty of Science, The Hong Kong Polytechnic University, Hong Kong, Hong Kong SAR, China, ⁴College of Osteopathic Medicine, Kansas City University, Kansas City, MO, United States

Background: Policies dealing with the Coronavirus Disease 2019 (COVID-19) pandemic vary across the globe, the different governmental responses then affect the public perception of COVID-19. Many unofficial Chinese media outlets frequently spread misinformation about COVID-19 and exaggerated reports of rare sequelae of Omicron for monetization and attention seeking, leading to panics in the Chinese public. In comparison the attitudes toward Omicron in other countries around the world, especially in North America and Western Europe have shifted to a more relaxed stance.

Objective: This article primarily aims to investigate the association between Chinese people's attitudes toward the potential after-effects of Omicron and their anxiety status, as compared to these of people living in North America or Western Europe.

Methods: We conducted a questionnaire survey *via* the Credamo and collected valid data from 500 Chinese (not living in Shanghai), another 500 Chinese (living in Shanghai) and 500 people living in North America or Western Europe in June 2022. Kendall's coefficient of rank correlation tau-sub-b was used to examine this association.

Results: The results suggested that subjective attitudes of Chinese participants toward the sequelae of Omicron were positively and significantly associated with their anxiety status [i.e., the Generalized Anxiety Disorder 7-item (GAD-7) scores] in Shanghai (China) (Tb = 0.44, p < 0.01) and other parts of China outside Shanghai (Tb = 0.37, p < 0.01). However, no such significant correlation was found in North America & Western Europe (Tb = -0.01, p > 0.05).

Conclusion: Our findings showed that Chinese participants who were more worried about the after-effects of Omicron had higher levels of anxiety. Although it is true that Long COVID-19 should be a concern, exaggerated media reporting can impact negatively on an individual's mental wellbeing. Only through the dissemination of robust scientific studies, the misinformation and the fears that follow it can be put to rest.

KEYWORDS

anxiety, COVID-19, Omicron, pandemic, sequelae

Introduction

Background

Coronavirus Disease 2019 (COVID-19) outbreak elicited by severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) began in December 2019 (1). As of June 7, 2022, the COVID-19 has led to more than 536 million confirmed cases and 6.32 million deaths globally (2). It has been considered a serious event, impacting significantly on the mental health of the global population (3).

Coronavirus disease 2019 and anxiety

In order to reduce the risk of coronavirus exposure in public, staying-at-home campaign was highly recommended or even mandated (4). As a result, the fear of contracting the virus, high unemployment due to economic loss, interrupted daily routine during recurrent periods of lockdown, the inability of engaging in most canceled outdoor events and other factors induced by COVID-19 severely impacted public mental health (5). In the general Chinese population, varying degrees of anxiety resulted from many factors, such as overestimating threat and intolerance of uncertainty to COVID-19, ranging from 2 to 37%, yielded a pooled prevalence of 19.1% (6–14). In North America and Western Europe, a pooled prevalence of anxiety was slightly lower than 15% (15).

Omicron and after-effects

Omicron, a newest and most popular variant of the Coronavirus, firstly discovered on November 24, 2021, has clinical characteristics mainly consisting of mild symptoms but extremely high communicable capacity (16). With reference to the after-effects of COVID-19, previous studies showed that delta variant or other preceding variants could possibly cause patients many impacts such as hair loss, altered sense of smell and taste (17–19), while the consequence of Omicron is unclear and still under evaluation (20).

Omicron in China

The COVID-19 pandemic was well controlled in China owing to its zero-tolerance approach to coronavirus applied in the past 2 years (21-24), but for this reason, no herd immunity barrier has been established (25); meanwhile, other countries (especially Western countries) attempted coexistence with COVID-19 in order to return things that were severely impacted such as economy by pandemic to normal (26). Hence, once the pandemic spread internationally, the potential risk caused by highly contagious Omicron to the whole country (i.e., China) could be very high (27). Unfortunately, Omicron suddenly broke out in Shanghai, China starting in March and continued to grow at a rate of about 10,000 confirmed patients per day until May (28, 29). The dire situation was not brought under control until early June (30, 31).

People's perception of coronavirus disease 2019 *via* media outlets in China and the West

It is important to note that all traditional news media outlets in China are controlled by Central Publicity Department (CPD) the information published come under more censorship than their western counterparts (32). As is often the case, alternative forms of traditional media flourished instead under the radar of the government control regime, citizen journalism as it is coined became the new way many people obtain news (33, 34). From this understanding it is evident that the flourishing unofficial Chinese media outlets, mostly owned by individuals and private companies, are comparable with traditional news media in the West in terms of function. The popular hosting platforms (e.g., WeChat, Sina Microblog, ZhiHu, and Bilibili) for the unofficial Chinese media outlets are also in and of itself a social media platform, making it very easy to share articles and comments to friends and families. Similar to how western traditional media also uses social media (e.g., Twitter and Facebook) to promote their news articles for views. Therefore, from this perspective, the comparison of how media affects individuals' perception of events is valid in this context, though exceptions that disputable opinions or comments are restricted to access may still exist in these Chinese media outlets.

However, compared to North America and Western Europe, where more than half of the residents there have had actual experience with COVID-19 (35, 36), the low COVID-19 prevalence in China led to a greater likelihood that Chinese people obtain the information of COVID-19 through the media outlets (37, 38). Hence, Chinese people's perception toward COVID-19 could be, to a much larger extent, dependent on unofficial reports of these media outlets, which have been found an effective medium to acquire relevant information for the public (37–39).

The current study

Many Chinese owned media outlets frequently spread non-evidence-based information of COVID-19 or greatly exaggerated rare sequelae of Omicron lacking common consent of systematic study for the sake of attention, leading Chinese people to panic situation, in comparison with the large shift in attitudes toward Omicron in other countries around the world, especially in North America and Western Europe (40-42). Therefore, this article mainly intends to explore the association between Chinese people's attitudes toward the potential sequelae of Omicron and their anxiety levels, as compared to these of people residing in North America or Western Europe. In addition, this study is also intended to present up-to-date information regarding risks of evidence-based sequelae that Omicron may cause to patients. Based on the backgrounds of COVID-19 and its information's propagation via media outlets mentioned above in these countries, we propose Hypothesis 1 and Hypothesis 2.

- 1. Chinese people holding more negative attitudes toward the after-effects of Omicron will also have higher anxiety scores, compared to those residing in North America and Western Europe.
- 2. Residents in North America and Western Europe will have lower anxiety levels in terms of Generalized Anxiety Disorder 7-item (GAD-7) mean scores, compared to those residing in China.

Materials and methods

Overview

We conducted a questionnaire survey via the Credamo, a professional online survey platform similar to Qualtrics Online Sample (43), by randomly recruiting intending participants who were interested in our study, starting on June 1, 2022, and ending on June 8, 2022, and the use of human data from the surveys was carried out ethically in accordance with the principles of the Declaration of Helsinki (as revised in 2013). During this process, a Chinese version of questionnaire was used to collect the data from Chinese participants directly through a webpage-based answering platform on Credamo.¹ An English version of the same questionnaire was separately submitted to the Credamo company to help collect the data from North America and Western Europe. To eliminate any potential misunderstanding of participants to questions due to different versions of questionnaires (i.e., Chinese vs. English), each question in the questionnaire was followed with a relevant example explaining the intention we were hoping to ask. On the first page of the questionnaire survey all participants received an adequate description of the purpose of the survey and were asked to tick a box to confirm an online informed consent prior to filling out survey. Furthermore, all data were collected anonymously through the Credamo using continuous identifier numbers to distinguish participants instead of recording their names or other sensitive information.

For survey answering quality, two attention check questions at different point in the survey were used. A one US dollar or RMB/GBP/Euro equivalent monetary incentive was offered for each participant who completed the survey. Meanwhile, we manually checked the time taken for completing each survey as well as the IP address of responders in case of the same responders joining the survey multiple times. Moreover, on the first page of the questionnaire survey, participants were informed about finishing the questionnaire truthfully under personal anxiety status developed explicitly during the pandemic era that Omicron dominated. Also, they were strictly informed that only those who did not experience any personal COVID-19 related situation that had caused their mental status deteriorated severely, were permitted to complete the questionnaire survey.

Questionnaire contents

The questionnaire was mainly comprised of the following information collected:

¹ www.credamo.com

- 1. Demographic information
- 2. How many shots of vaccine did you get?
- 3. Have you ever been infected with COVID-19?
- 4. Do you support coexistence or zero-tolerance approach of the Omicron-dominated pandemic in your country?
- 5. Do you have psychological fear toward your real-life friends who were infected with COVID-19 (i.e., do you want to be wary of them inwardly)?
- Subjective attitudes toward Omicron about its after-effects (i.e., "to what extent do you think Omicron could cause sequelae?")
- 7. The 7-item Generalized Anxiety Disorder-7 (GAD-7) scale.

A total of 1,500 people were initially recruited to complete the questionnaire survey. Invalid data were excluded, and new participants were recruited until 1,500 individual data fulfilled our standard inclusion criteria. Finally, a total of 1,623 people living in China, North America or Western Europe were independently recruited and surveyed through the Credamo platform. Of them, 78 were excluded for the failure in the attention check questions (e.g., responded wrongly to the instruction "please chose the answer Blue"), 33 were excluded for completing the survey in less than 100 s, 12 were excluded with additional analyses for other reasons such as answering the questionnaire questions inconsistently or contradictorily. Eventually, a valid sample of 1,500 participants was analyzed collectively (688 females and 812 males; mean age = 26.74 years, SD = 3.81 years; age range: 18–34 years).

Generalized anxiety disorder 7-item scale

The anxiety status of the participants was assessed using the 7-item version of the Generalized Anxiety Disorder scale or GAD-7. It consists of seven items based on seven core symptoms, asking respondents how often they experienced these symptoms in the past 2 weeks, and is preferably used to measure an individual's proximate level of anxiety in a timely manner during the pandemic era (44-48). For each item, participants were asked to choose the degree to which they agreed or disagreed with the statement, on a scale of 0-3, with 0 denoting "not at all," 1 denoting "several days," 2 denoting "more than half the days" and 3 denoting "nearly every day." In the GAD-7 scale, total score of participants for the seven items ranging from 0 to 21 was summed up. A total score of 0-4, 5-9, 10-14, and 15-21 were classified as minimal anxiety, mild anxiety, moderate anxiety, and severe anxiety, respectively. Hence, a higher total score indicated a higher level of anxiety status of participants. In this study, the Cronbach's alpha coefficient for the total scale was 0.91, suggesting excellent overall internal consistency.

Statistical analysis

All statistical analyses were performed using the software program SPSS (version 26.0) except for the data cleaning process which included detection and removal of invalid or missing data completed on the Credamo data platform. A reliability test was conducted for the GAD-7 scale, using Cronbach's alpha coefficients as a measure of internal consistency ($\alpha > 0.70$ regarded as acceptable). Mean differences were compared by using parametric tests. Finally, Kendall's coefficient of rank correlation tau-sub-b was used to examine the association between the subjective attitudes of participants toward the aftereffects of Omicron (ordinal variable) and the GAD-7 self-report scale scores (continuous variable), according to Khamis (49).

Results

Sample characteristics

There were 1,623 individuals from mainland China (Shanghai vs. non-Shanghai), North America and Western Europe, who enrolled in the survey, and 1,500 (92.4%) were included in the analysis participants after data cleaning. Relevant descriptive statistics were presented in Table 1.

Mean comparison of general anxiety disorder 7-item scores

Regarding the mean differences of GAD-7 shown in **Table 2**, our results suggested that no any statistically significant difference was found in terms of gender, vaccination status, infection status of participants from outside Shanghai, view of coexistence with COVID-19, psychological fear toward friends infected with COVID-19 within groups (i.e., Non-Shanghai area (China), Shanghai (China), and North America and Western Europe; all p > 0.05), except for subjective attitudes of participants toward the sequelae of Omicron (all p < 0.05). However, there was strongly significant difference with reference to the mean of GAD-7 scores by area between groups as shown in **Figure 1** (mean \pm *SD* = 5.768 \pm 3.59, 9.034 \pm 3.93 and 3.94 \pm 2.53, respectively; *F* = 287.485, *p* < 0.001).

Kendall's tau-b analysis model

In **Table 3**, Kendall's coefficient of rank correlation tau-subb as a type of inferential statistics was conducted to investigate the correlation between subjective attitudes of participants toward the sequelae of Omicron and their anxiety levels among different areas. Most importantly, it was found that subjective attitudes of Chinese participants toward the sequelae

n (%)

<i>N</i> = 1,500	China (Shanghai) n = 500	China (non-Shanghai) n = 500	North America or Europe n = 500			
Variables						
Gender						
Male	278 (55.6)	291 (58.2)	243 (48.6)			
Female	222 (44.4)	209 (41.8)	257 (51.4)			
Vaccination status						
1 dose	13 (2.6)	10 (2)	68 (13.6)			
2 doses	28 (5.6)	110 (22)	105 (21.0)			
3 doses	459 (91.8)	380 (76)	327 (65.4)			
Infection status						
Yes	14 (2.8)	4 (0.8)	393 (78.6)			
No	486 (97.2)	496 (99.2)	107 (21.4)			
View of coexistence with COVID-19						
Support zero-tolerance approach (because of fearing sequelae of Omicron)	212 (42.4)	335 (67.0)	39 (7.8)			
Support zero-tolerance approach (because of misgiving medical resource crowding)	93 (18.6)	79 (15.8)	20 (4)			
Support co-existence with virus as much as possible	195 (39)	86 (17.2)	441 (88.2)			
Psychological fear toward friends infected with COVID-19						
No	247 (49.4)	259 (51.8)	483 (96.6)			
Yes	253 (50.6)	241 (48.2)	17 (3.4)			
Subjective attitude toward the sequelae of Omicron						
No sequelae	5 (1.0)	11 (2.2)	283 (56.6)			
Mild sequelae	54 (10.8)	229 (45.8)	106 (21.2)			
Moderate sequelae	185 (37.0)	146 (29.2)	67 (13.4)			
Severe sequelae	256 (51.2)	114 (22.8)	44 (8.8)			

of Omicron were positively and significantly associated with their anxiety status (i.e., GAD-7 scores) in Shanghai (China) (Tb = 0.44, p < 0.01) and other parts of China outside Shanghai (Tb = 0.37, p < 0.01). However, no such significant correlation was found in North America & Western Europe (Tb = -0.01, p > 0.05). This meant that there was a tendency for Chinese participants to report higher levels of anxiety the more they feared the sequelae of Omicron.

Discussion

The given study mainly examined the association between participants' attitudes toward the potential after-effects of Omicron and their anxiety status, and mean of GAD-7 scores of participants from different areas (Shanghai vs. Outside Shanghai within China vs. North America and Western Europe). The results of our study primarily showed that the more severe the sequelae of omicron the Chinese participants perceived, the higher their anxiety levels, but such relationship did not statistically and significantly exist in participants from North America and Western Europe. Also, regarding the mean of GAD-7 scores among different areas, participants from North America and Western Europe had relatively lowest anxiety levels (mean \pm SD = 3.94 \pm 2.53), followed by participants from China outside Shanghai (mean \pm SD = 5.768 \pm 3.59), and then participants from Shanghai (mean \pm *SD* = 9.034 \pm 3.93). These findings were consistent with our primary hypotheses in the introduction section. In addition, according to the results of one-way ANOVA as shown in Table 2, no statistical and significant difference of participants' levels of anxiety was found in terms of gender, vaccination status, infection status of participants from outside Shanghai, subjective view of coexistence with COVID-19, personal psychological fear toward friends infected with COVID-19. However, there was a significant difference in terms of the infection status (Yes vs. No) of participants in Shanghai (China) and their corresponding GAD-7 mean scores. This might be understandable that the people in Shanghai were urgently required to respond a sudden pandemic situation, which led to a panic to the public with increased anxiety.

TABLE 2 General anxiety disorder 7-item (GAD-7) scores of participants (N = 1,500).

Variables

GAD-7 scores

	Means (SD)		[95%CI]		p				
	CN SH	CN Non-SH	NA and WE	CN SH	CN Non-SH	NA and WE	CN SH	CN Non-SH	NA and WE
Gender							0.578	0.731	0.664
Male	8.74 (3.97)	5.23 (2.34)	3.77 (1.80)	[8.27, 9.21]	[4.96, 5.50]	[3.54, 4.00]			
Female	9.11 (3.75)	6.03 (2.08)	4.11 (2.66)	[8.62, 9.60]	[5.75, 6.31]	[3.78, 4.44]			
Vaccination status							0.344	0.544	0.650
1 Dose	8.69 (3.97)	7.00 (3.33)	4.21 (2.59)	[6.53, 10.85]	[4.93, 9.07]	[3.59, 4.83]			
2 Doses	10.07 (4.81)	5.70 (3.71)	3.92 (2.57)	[8.29, 11.85]	[5.01, 6.39]	[3.43, 4.41]			
3 Doses	8.98 (3.87)	5.76 (3.56)	3.89 (2.52)	[8.63, 9.33]	[5.40, 6.12]	[3.62, 4.16]			
Infection status							0.000	0.837	0.067
Yes	4.06 (2.65)	10.50 (4.04)	3.89 (2.48)	[2.67, 5.45]	[6.54, 14.46]	[3.64, 4.14]			
No	7.07 (4.11)	5.73 (3.56)	4.15 (2.74)	[6.70, 7.44]	[5.42, 6.04]	[3.63, 4.67]			
VOC							0.936	0.092	0.099
SZAFS	9.17 (3.86)	5.63 (3.54)	4.73 (3.07)	[8.62, 9.72]	[5.25, 6.01]	[3.77, 5.69]			
SZAMM	8.93 (3.93)	7.16 (4.17)	4.64 (2.29)	[8.13, 9.73]	[6.24, 8.08]	[3.64, 5.64]			
SCV	9.35 (3.31)	6.05 (3.17)	4.50 (2.87)	[8.89, 9.81]	[5.38, 6.72]	[4.23, 4.77]			
PFTF							0.823	0.526	0.284
Yes	9.13 (3.86)	5.83 (3.54)	3.29 (2.14)	[8.65, 9.61]	[5.38, 6.28]	[2.27, 4.31]			
No	8.94 (4.00)	5.71 (3.64)	3.96 (2.55)	[8.44, 9.44]	[5.27, 6.15]	[3.73, 4.19]			
SATSO							0.000	0.000	0.060
No sequelae	6.80 (5.36)	3.64 (3.91)	3.95 (2.50)	[2.11, 11.49]	[1.33, 5.95]	[3.66, 4.24]			
Mild sequelae	7.04 (3.93)	4.64 (3.36)	3.45 (2.69)	[5.99, 8.09]	[4.20, 5.08]	[2.94, 3.96]			
Moderate sequelae	7.17 (2.73)	5.31 (2.34)	4.42 (2.23)	[6.78, 7.56]	[4.93, 5.69]	[3.89, 4.95]			
Severe sequelae	10.85 (3.79)	8.82 (3.61)	4.34 (2.68)	[10.39, 11.31]	[8.16, 9.48]	[3.55, 5.13]			

CN, China, SH, Shanghai, NA, North America; WE, Western Europe; VOC, View of coexistence with COVID-19; SZAFS, Support zero-tolerance approach (because of fearing sequelae of Omicron); SZAMM, Support zero-tolerance approach (because of misgiving medical resource crowding); SCV, Support co-existence with virus as much as possible; PFTF, Psychological fear toward friends infected with COVID-19; SATSO, Subjective attitude toward the sequelae of Omicron.

Shan et al.



TABLE 3 Correlation between SATSO and GAD-7 scores among different areas.

Kendall's tau	ı-b	SATSO in China (Shanghai) (<i>n</i> = 500)	SATSO in China outside Shanghai (<i>n</i> = 500)	SATSO in North America and Western Europe $(n = 500)$
GAD-7 scores	Correlation coefficient	0.44**	0.37**	-0.01
	Sig. (2-tailed)	0.00	0.00	0.70

N = 1,500. **Representing the correlation is significant at the 0.01 level (p < 0.01).

SATSO, Subjective attitude toward the sequelae of Omicron.

Admittedly, participants' mindsets toward the pandemic might differ due to different cultures (50). But in the context of the COVID-19 pandemic in today's advanced technological society, media outlets have been seen as useful means of spreading information about COVID-19 and measuring public attention toward COVID-19 in both China and the Western countries (37-39, 51, 52). Online COVID-19 infodemic (i.e., pandemic of misinformation), without prudent journalistic judgments of media content, could be easily and quickly disseminated and thus influence public opinions (39), therefore resulting in deadly consequences (51, 52). In addition, as we mentioned in the Introduction section, Chinese people's perceptions toward COVID-19 could largely rely on the propagation of information of COVID-19 through media outlets, in comparison with the residents in North America and Western Europe, where a virus co-existence policy with relatively few restrictions to the public resulted in a great number of people being affected with COVID-19; but meanwhile, these people were thus allowed to have an actual experience of how long-term COVID-19 impacts their body, rather than only acquiring relevant information *via* media reports. Hence, the propagation of information about COVID-19 should be concerned, especially for the Chinese public.

Over the course of the COVID-19 pandemic, researchers are still struggling with the after-effects of coronavirus as it continued to evolve. Nowadays, as the COVID-19 variants prior to Omicron have nearly fade away, investigating and discussing the potential sequelae of Omicron that is the most prevalent variant in the current pandemic situation is necessary (53). Nevertheless, due to the significant lag in the publication of studies relevant to COVID-19 sequelae, the findings of recently published articles may not be applied to the latest Omicron situation. For example, the study suggesting that COVID-19 could lead to greater reduction in brain gray matter thickness was conducted in 2021 when the participants involved in this study were infected with the earliest variant of COVID-19 rather than Omicron (54); meanwhile, these participants were unvaccinated and generally older. Therefore, it is difficult to match these sequelae with the current less threatening Omicron. But a very recently published article suggested that the

probability of Omicron causing long-term impacts to patients (4.5% of Omicron patients developed sequelae) was half that of Delta (10.8%) (55).

Some anecdotal findings reported by Chinese mass media indicated that most negative impacts of Omicron sequelae to human body were not reversible. However, that was not what previous studies actually found. For example, Zhao et al. suggested that mild Omicron sequelae such as slightly reduced attention and memory ability, which were even not perceived by participants themselves, were much improved over time (56). Similarly, another study followed the health status of patients with COVID-19 in Wuhan for 1 year after discharge from the hospital, and found that the after-effects of COVID-19 such as fatigue, sleeping disturbance and depression initially presented were improved greatly over time in these patients (57). Moreover, given that the study was conducted on the first batch of patients infected with COVID-19 in Wuhan, its findings were also not time-sensitive in the current context of Omicron.

Many anecdotal news online also stated that there was evidence that COVID-19 could induce male impotence. In fact, although a relevant study did suggest that COVID-19 may induce testicular damage, which could eventually result in decreased libido and fertility, the subjects involved in this study were animal rather than human patients; meanwhile, it was found that such negative impacts could be preventable by vaccination (58). However, when the findings of this study were reported by the mass media outlets, they overly exaggerated the impacts of COVID-19 by just saying "New study shows COVID-19 infection could cause testicular atrophy and reduced fertility in men." The lack of evidence for statements such as the effects of COVID-19 on fertility and intelligence is not only unfair and discriminatory to those infected with COVID-19, but also affects the psychological state of those who have never suffered COVID-19 infection and increases their anxiety level. Therefore, the mass media reports were misleading to the public, which should have been avoided as much as possible. Regarding the effectiveness of vaccination against long COVID-19, two studies by Ayoubkhani et al. demonstrated that people who completed two doses of vaccine were less likely to develop long-term sequelae after being infected with COVID-19 (59, 60).

Furthermore, it is also important to note that the COVID-19 sequelae are not necessarily related to the COVID-19 itself. More specifically, any influenza or infectious disease may also induce similar negative impacts as COVID-19. For instance, a cross-sectional study with a large French cohort suggested that the so-called sequelae of COVID-19 perceived by patients themselves may be more psychological or actually caused by other diseases than the laboratory-confirmed result of COVID-19 infection (61).

Another issue to note is that our study found around half of the Chinese participants (Table 1) having psychological fear toward friends infected with COVID-19, though no significant difference between such mindsets and their anxiety levels was found in terms of GAD-7 mean scores. Therefore, we should advocate avoiding whether implicitly or explicitly discriminating people infected with COVID-19 who have the potential to suffer from various degrees of psychological disorders due to surrounding pressures such as social rejection.

Overall, with the widespread vaccination around the world, threats of the COVID-19 pandemic have been weakened. As can be seen from the outbreak in Shanghai recently, a large number of asymptomatic patients, even confirmed cases, were mainly mild symptom patients. Given that the global pandemic has become the norm, a total zero-out policy is not desirable. What we should do is to face the COVID-19 bravely with a more open and inclusive mind. In the current article, it seems that the fear of Omicron after-effects is more frightening than the COVID-19 itself in Chinese population. Thus, policy makers should enhance the public's awareness of the latest change of pandemic situation, to eliminate unnecessary worries and reduce the psychological burden of Chinese people.

Limitation

The current study has several limitations. First, this study was a cross-sectional study that might restrict causal inference. Second, the sample size was not large enough, thereby limiting the generalizability of this study. Third, this study was based on self-reported responses of participants. Although the data derived from an online professional data collection platform, more study with more professional research methods in similar topics is needed to carry out in the future, when conditions are permitted. Finally, another limitation in this study is the fact that participants' media exposure was hard to track and measure, so a direct correlation between participants' perceptions toward COVID-19 or Omicron specifically resulted from exposure to media outlets and their anxiety levels could be biased and still needs more study to further demonstrate.

Conclusion

Currently, the global pandemic is subsiding as the novel coronavirus gradually evolves in a less harmful direction. However, due to the exaggeration about the long effects of Omicron by mass media outlets, which is currently the most prevalent variant of COVID-19, a variety of fears about Omicron long effects and a great deal of unpredictability about the future pandemic continue to plague people. In the current study we found that Chinese participants who were more worried about the after-effects of Omicron had higher levels of anxiety. Overall, although we still need to pay sufficient attention to COVID-19 and its long effects, we should take everything related to COVID-19 seriously based on the available scientific evidence, and not easily believe exaggerated or even false reports in the mass media. Also, to eliminate unnecessary worries and reduce the psychological burden of Chinese people, policy makers should put sufficient efforts to enhance the public's awareness of the latest change of pandemic situation. In the future, more relevant studies are needed to reveal the long-term impacts of Omicron or subsequent variants of COVID-19.

Data availability statement

The raw data supporting the conclusions of this article will be made available by the authors, without undue reservation.

Ethics statement

Ethical review and approval were not required for this study on human participants in accordance with the local legislation and institutional requirements. The participants were notified to provide their written informed consent online before participation in this study.

Author contributions

DS proposed the research idea, collected and analyzed the data from China, and wrote the initial manuscript. CL performed the literature search, provided the idea for the analysis, and collected the data from North America and

References

1. Huang C, Wang Y, Li X, Ren L, Zhao J, Hu Y, et al. Clinical features of patients infected with 2019 novel coronavirus in Wuhan, China. *Lancet.* (2020) 395:497-506. doi: 10.1016/S0140-6736(20)30183-5

2. Elflein J. Number of coronavirus (COVID-19) cases, recoveries, and deaths worldwide. Hamburg: Statista (2022).

3. Rauch SA, Simon NM, Rothbaum BO. Rising tide: Responding to the mental health impact of the COVID-19 pandemic. *Depress Anxiety.* (2020) 37:505-9. doi: 10.1002/da.23058

4. Quiroga B, Giorgi M, Barril G. During COVID-19 stay at home even if you are a hemodialysis patient. *Ther Apher Dial.* (2021) 25:705-6. doi: 10.1111/1744-9987.13615

5. Lakhan R, Agrawal A, Sharma M. Prevalence of depression, anxiety, and stress during COVID-19 pandemic. *J Neurosci Rural Pract.* (2020) 11:519–25. doi: 10.1055/s-0040-1716442

6. Ahmed MZ, Ahmed O, Aibao Z, Hanbin S, Siyu L, Ahmad A. Epidemic of COVID-19 in China and associated psychological problems. *Asian J Psychiatry.* (2020) 51:102092. doi: 10.1016/j.ajp.2020.102092

7. Hamidi S, Zandiatashbar A. Compact development and adherence to stay-athome order during the COVID-19 pandemic: a longitudinal investigation in the United States. *Landsc Urban Plan.* (2021) 205:103952.

8. Huang Y, Zhao N. Generalized anxiety disorder, depressive symptoms and sleep quality during COVID-19 outbreak in China: A web-based cross-sectional survey. *Psychiatry Res.* (2020) 288:112954. doi: 10.1016/j.psychres.2020. 112954

Western Europe. SYL contributed to the substantial revisions of the manuscript in terms of reviewers' comments. YDZ checked the accuracy of the data analysis and was responsible for final proofreading and contributing to minor revisions of the latest manuscript. All authors have read and approved the final version of the manuscript and agreed with the order of the presentation of the authors.

Acknowledgments

We thank the professional online survey platform Credamo for data provided in this article.

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

9. Lei L, Huang X, Zhang S, Yang J, Yang L, Xu M. Comparison of prevalence and associated factors of anxiety and depression among people affected by versus people unaffected by quarantine during the COVID-19 epidemic in Southwestern China. *Med Sci Monit.* (2020) 26:e924609. doi: 10.12659/MSM.924609

10. Pappa S, Ntella V, Giannakas T, Giannakoulis VG, Papoutsi E, Katsaounou P. Prevalence of depression, anxiety, and insomnia among healthcare workers during the COVID-19 pandemic: A systematic review and meta-analysis. *Brain Behav Immunity.* (2020) 88:901–7. doi: 10.1016/j.bbi.2020.05.026

11. Qiu J, Shen B, Zhao M, Wang Z, Xie B, Xu Y. A nationwide survey of psychological distress among Chinese people in the COVID-19 epidemic: Implications and policy recommendations. *Gen Psychiatry*. (2020) 33:e100213. doi: 10.1136/gpsych-2020-100213

12. Urairak B. Effect of the COVID-19 pandemic related mental health on state anxiety in Thailand. *Asian Adm Manage Rev.* (2022) 5(1):1–6.

13. Wang C, Pan R, Wan X, Tan Y, Xu L, Ho CS, et al. Immediate psychological responses and associated factors during the initial stage of the 2019 coronavirus disease (COVID-19) epidemic among the general population in China. *Int J Environ Res Public Health*. (2020) 17:1729. doi: 10.3390/ijerph1705 1729

14. Zhou SJ, Zhang LG, Wang LL, Guo ZC, Wang JQ, Chen JC, et al. Prevalence and socio-demographic correlates of psychological health problems in Chinese adolescents during the outbreak of COVID-19. *Eur Child Adolesc Psychiatry*. (2020) 29:749–58. doi: 10.1007/s00787-020-01541-4

15. Li Y, Scherer N, Felix L, Kuper H. Prevalence of depression, anxiety and posttraumatic stress disorder in health care workers during the COVID-19 pandemic: A systematic review and meta-analysis. PLoS One. (2021) 16:e0246454. doi: 10.1371/ journal.pone.0246454

16. Meo SA, Meo AS, Al-Jassir FF, Klonoff DC. Omicron SARS-CoV-2 new variant: Global prevalence and biological and clinical characteristics. *Eur Rev Med Pharmacol Sci.* (2021) 25:8012–8.

17. Hussain N, Agarwala P, Iqbal K, Omar HMS, Jangid G, Patel V, et al. A systematic review of acute telogen effluvium, a harrowing post–COVID–19 manifestation. *J Med Virol.* (2022) 94:1391–401. doi: 10.1002/jmv.27534

18. Izquierdo-Dominguez A, Rojas-Lechuga MJ, Mullol J, Alobid I. Olfactory dysfunction in the COVID-19 outbreak. *J Investig Allergol Clin Immunol.* (2020) 30:317–26. doi: 10.18176/jiaci.0567

19. Mastrangelo A, Bonato M, Cinque P. Smell and taste disorders in COVID-19: From pathogenesis to clinical features and outcomes. *Neurosci Lett.* (2021) 748:135694. doi: 10.1016/j.neulet.2021.135694

20. Lapid N. Explainer: Why you should still try to avoid catching omicron. London: Reuters (2022).

21. Burki T. China's successful control of COVID-19. Lancet Infect Dis. (2020) 20:1240-1. doi: 10.1016/S1473-3099(20)30800-8

22. Chen J, Guo X, Pan H, Zhong S. What determines city's resilience against epidemic outbreak: Evidence from China's COVID-19 experience. *Sustain Cities Soc.* (2021) 70:102892. doi: 10.1016/j.scs.2021. 102892

23. Qi J, Zhang D, Zhang X, Takana T, Pan Y, Yin P, et al. Short and medium-term impacts of strict anti-contagion policies on non-COVID-19 mortality in China. *Nat Hum Behav.* (2022) 6:55–63. doi: 10.1038/s41562-021-01189-3

24. Silver A. Covid-19: Why China is sticking to "zero tolerance" public health measures. *BMJ.* (2021) 375:n2756. doi: 10.1136/bmj.n2756

25. An Z, Wang F, Pan A, Yin Z, Rodewald L, Feng Z. Vaccination strategy and challenges for consolidating successful containment of COVID-19 with population immunity in China. *BMJ*. (2021) 375:e066125. doi: 10.1136/bmj-2021-066125

26. Li KLA. Success and challenges: China's state capacity of popular mobilization in the COVID-19 pandemic. *Chin Stud.* (2022) 11:68–78. doi: 10.4236/chnstd.2022. 112006

27. Mallapaty S. China's Zero-COVID strategy: What happens next? Nature. (2022) 602:15-6. doi: 10.1038/d41586-022-00191-7

28. Baum SD, Adams VM. Pandemic refuges: lessons from 2 years of COVID-19. *Risk Anal.* (2022):1–9. doi: 10.1111/risa.13953

29. Bradsher K. Surge of omicron infections prompts lockdowns in China. New York, NY: The New York Times (2022).

30. Wang EF. Shanghai starts coming back to life as COVID lockdown eases. New York, NY: AP News (2022).

31. Pike L, Nagorski T. Shanghai's COVID lockdown is over. Its economic problems aren't. Grid. (2022). Available online at: https://www.grid.news/story/global/2022/06/09/shanghais-covid-lockdown-is-over-its-economic-problems-arent/(accessed June 9, 2022)

32. Hassid J. Controlling the Chinese media: An uncertain business. Asian Survey. (2008) 48:414-30. doi: 10.1525/as.2008.48.3.414

33. Xia S. A study on the profit model of we media in China. *Glob Med J.* (2017) 15:28.

34. Xin X. The impact of "citizen journalism" on Chinese media and society. J Pract. (2010) 4:333-44. doi: 10.1080/17512781003642931

35. Mandavilli A. *The coronavirus has infected more than half of Americans, the C.D.C. Reports.* New York, NY: The New York Times (2022).

36. Daria S, Islam MR. The SARS—CoV—2 omicron wave is indicating the end of the pandemic phase but the COVID—19 will continue. *J Med Virol.* (2022) 94:2343. doi: 10.1002/jmv.27635

37. Lin Y, Hu Z, Alias H, Wong LP. Influence of mass and social media on psychobehavioral responses among medical students during the downward trend of COVID-19 in Fujian, China: Cross-sectional study. *J Med Internet Res.* (2020) 22:e19982. doi: 10.2196/19982

38. Zhao Y, Cheng S, Yu X, Xu H. Chinese public's attention to the COVID-19 epidemic on social media: Observational descriptive study. *J Med Internet Res.* (2020) 22:e18825. doi: 10.2196/18825

39. Luo Y, Cheng Y. The presumed influence of COVID-19 misinformation on social media: Survey research from two countries in the global health crisis. *Int J Environ Res Public Health.* (2021) 18:5505. doi: 10.3390/ijerph18115505

40. Schmelz K, Bowles S. Opposition to voluntary and mandated COVID-19 vaccination as a dynamic process: Evidence and policy implications of changing

beliefs. Proc Natl Acad Sci. (2022) 119:e2118721119. doi: 10.1073/pnas.211872 1119

41. Tang JW, Caniza MA, Dinn M, Dwyer DE, Heraud JM, Jennings LC, et al. An exploration of the political, social, economic and cultural factors affecting how different global regions initially reacted to the COVID-19 pandemic. *Interface Focus.* (2022) 12:20210079. doi: 10.1098/rsfs.2021.0079

42. Toovey OT, Tang JW. The emergence of the omicron variant. *Clin Infect Pract.* (2022) 13:100134. doi: 10.1016/j.clinpr.2022.100134

43. Gong S, Lu JG, Schaubroeck JM, Li Q, Zhou Q, Qian X. Polluted psyche: Is the effect of air pollution on unethical behavior more physiological or psychological? *Psychol Sci.* (2020) 31:1040–7. doi: 10.1177/0956797620943835

44. Löwe B, Decker O, Müller S, Brähler E, Schellberg D, Herzog W, et al. Validation and standardization of the generalized anxiety disorder screener (GAD-7) in the general population. *Med Care.* (2008) 46:266–74. doi: 10.1097/MLR. 0b013e318160d093

45. Singh P, Cumberland WG, Ugarte D, Bruckner TA, Young SD. Association between generalized anxiety disorder scores and online activity among US adults during the COVID-19 pandemic: Cross-sectional analysis. *J Med Internet Res.* (2020) 22:e21490. doi: 10.2196/21490

46. Spitzer RL, Kroenke K, Williams JB, Löwe B. A brief measure for assessing generalized anxiety disorder: The GAD-7. *Arch Internal Med.* (2006) 166:1092–7. doi: 10.1001/archinte.166.10.1092

47. Toussaint A, Hüsing P, Gumz A, Wingenfeld K, Härter M, Schramm E, et al. Sensitivity to change and minimal clinically important difference of the 7item generalized anxiety disorder questionnaire (GAD-7). J Affect Disord. (2020) 265:395–401. doi: 10.1016/j.jad.2020.01.032

48. Williams N. The GAD-7 questionnaire. Occup Med. (2014) 64:224–224. doi: 10.1093/occmed/kqt161

49. Khamis H. Measures of association: How to choose? J Diagn Med Sonogr. (2008) 24:155–62. doi: 10.1177/8756479308317006

50. Lin F, Chen X, Cheng EW. Contextualized impacts of an infodemic on vaccine hesitancy: The moderating role of socioeconomic and cultural factors. *Inf Process Manag.* (2022) 59:103013. doi: 10.1016/j.ipm.2022.103013

51. Bridgman A, Merkley E, Loewen PJ, Owen T, Ruths D, Teichmann L, et al. The causes and consequences of COVID-19 misperceptions: understanding the role of news and social media. *Harv Kennedy Sch Misinformation Rev.* (2020) 1:1-18. doi: 10.37016/mr-2020-028

52. Gisondi MA, Barber R, Faust JS, Raja A, Strehlow MC, Westafer LM, et al. A deadly infodemic: Social media and the power of COVID-19 misinformation. *J Med Internet Res.* (2022) 24:e35552. doi: 10.2196/35552

53. Katella K. Omicron, Delta, alpha, and more: What to know about the coronavirus variants. New Haven CT: Yale Medicine (2022).

54. Douaud G, Lee S, Alfaro-Almagro F, Arthofer C, Wang C, McCarthy P, et al. SARS-CoV-2 is associated with changes in brain structure in UK biobank. *Nature.* (2022) 604:697–707. doi: 10.1038/s41586-022-04569-5

55. Antonelli M, Pujol JC, Spector TD, Ourselin S, Steves CJ. Risk of long COVID associated with delta versus omicron variants of SARS-CoV-2. *Lancet.* (2022) 399:2263–4. doi: 10.1016/S0140-6736(22)00941-2

56. Zhao S, Shibata K, Hellyer PJ, Trender W, Manohar S, Hampshire A, et al. Rapid vigilance and episodic memory decrements in COVID-19 survivors. *Brain Commun.* (2022) 4:fcab295. doi: 10.1093/braincomms/fcab295

57. Huang L, Yao Q, Gu X, Wang Q, Ren L, Wang Y, et al. 1-year outcomes in hospital survivors with COVID-19: A longitudinal cohort study. *Lancet.* (2021) 398:747–58. doi: 10.1016/S0140-6736(21)01755-4

58. Lai CC, Shih TP, Ko WC, Tang HJ, Hsueh PR. Severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) and coronavirus disease-2019 (COVID-19): The epidemic and the challenges. *Int J Antimicrob Agents.* (2020) 55:105924. doi: 10.1016/j.ijantimicag.2020.105924

59. Ayoubkhani D, Bermingham C, Pouwels KB, Glickman M, Nafilyan V, Zaccardi F, et al. Trajectory of long COVID symptoms after COVID-19 vaccination: Community-based cohort study. *BMJ.* (2022) 377:e069676. doi: 10.1136/bmj-2021-069676

60. Ayoubkhani D, Bosworth ML, King S, Pouwels KB, Glickman M, Nafilyan V, et al. Risk of long covid in people infected with SARS-CoV-2 after two doses of a COVID-19 vaccine: Community-based, matched cohort study. *medRxiv* [Preprint] (2022). doi: 10.1101/2022.02.23.22271388

61. Matta J, Wiernik E, Robineau O, Carrat F, Touvier M, Severi G, et al. Association of self-reported COVID-19 infection and SARS-CoV-2 serology test results with persistent physical symptoms among french adults during the COVID-19 pandemic. *JAMA Intern Med.* (2022) 182:19–25.